Author's response to reviews

Title: Zuo Jin Wan, reverses P-gp mediate Multidrug resistance by inhibiting the activation of PI3K/Akt/NF-kappaB pathway

Authors:

hua sui (syh0808@163.com)
shufang pan (42481993@qq.com)
xiaoling fu (59045733@qq.com)
baohui jin (jbhrachel@gmail.com)
xuan liu (lxtw@hotmail.com)
lihong zhou (zlhtcm@hotmail.com)
qing ji (ttt99118@hotmail.com)
fenggang hou (fghou555@hotmail.com)
jianfeng cai (jianfengcai@usf.edu)
huirong zhu (zhu_huirong@126.com)
jianlin ren (renjianlin666@126.com)
qi li (lzwf@hotmail.com)

Version: 2 Date: 21 January 2014

Author's response to reviews: see over
Cover Letter

Dear Editors:

Thanks for your letter and the editorial requirements. Based on these comments and suggestions, we have made careful and necessary modification and asked a professional editing service to correct our language on the original manuscript. Herein, I have made a point-by-point description of the changes.

We shall look forward to hearing from you at your earliest convenience.

Yours sincerely,

Qi Li (Corresponding Author)

Department of Medical Oncology, Shuguang Hospital, Shanghai University of Traditional Chinese Medicine Shanghai, 201203. China.

Telephone: +8621 20256533

E-mail: lzwf@hotmail.com
1. Please could you include within your methods section where you obtained Zuo Jin Wan from and please provide more detail on how you prepared the various dilutions of Zuo Jin Wan used in your study.

Response: We have added the herbs purchased and extracted method in the methods section, and added a figure of the result in HPLC chromatographs of Zuo Jin Wan in Supplementary Material.